Author's response to reviews

Title: Primary Nasopharyngeal non-Hodgkin lymphomas: a retrospective review of 26 morrocan patients.

Authors:

Wafa Allam (wafallam@yahoo.fr)
Nabil Ismaili (ismailinabil@yahoo.fr)
Sanaa Elmajjaoui (dr_majjaoui@yahoo.fr)
Brahim K Elgueddari (b.elgueddari@medramo.ac.ma)
Mohamed Ismaili (ismailih2000@yahoo.fr)
Hassan Errihani (h_errihani@yahoo.fr)

Version: 7 Date: 10 November 2009

Author's response to reviews: see over
Formatting changes to be made:
-----------------------------
Major revisions (we require the author to make these changes)

Author list: the list of authors in the manuscript should be written exactly as they are in the submission system, both in style and order. The preferred style is 'First name Initial Last name' (e.g. Joe F Bloggs)

Research: medicine - Abstract must be structured into Background, Methods, Results, Conclusions. Please remember to also update the abstract details on the submission page. The abstract in the submission system must match exactly the abstract in your manuscript.

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

Thank you